Author’s response to reviews

Title: IKZF1 Genetic variants rs4132601 and rs11978267 and Acute Lymphoblastic Leukemia Risk in Tunisian children: a case-control study

Authors:
sana Mahjoub (sonson0903@hotmail.com)
Vera Chayeb (vchaieb@yahoo.f)
Hedia Zitouni (hediaaztn@gmail.com)
Rabeb El Ghali (elghali.rabeb@gmail.com)
Haifa Regaieg (regaieghaifa@yahoo.fr)
Mahjoub Touhami (touhamimahjoub@gmail.com)

Version: 2 Date: 18 Jun 2019

Author’s response to reviews:

Dear Editor

Thank you for answer

we have met the requirements of the journal, the presentation of the manuscript corresponds to the requirements of your journal; result tables are attached to manuscripts, the declarations are attached at the bottom of the manuscript.
Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgments

Sincerely